Author's response to reviews

Title: Capsule impaction presenting as acute small bowel perforation: two cases report.

Authors:

Giovanni D. De Palma (giovanni.depalma@unina.it)
Stefania Masone (stefaniamasone@yahoo.it)
Marcello Persico (marcellopersico@hotmail.com)
Saverio Siciliano (sicils@tin.it)
Francesca Salvatori (fri_salvatori@hotmail.com)
Francesco Maione (maj_one_doc@libero.it)
Dario Esposito (espodario83@gmail.com)
Giovanni Persico (giopersi@unina.it)

Version: 6 Date: 14 March 2012

Author's response to reviews:

title was changed
abstract was changed